Author's response to reviews

Title: Exhaled 8-isoprostane as a prognostic marker in sarcoidosis. A short term follow-up.

Authors:

Wojciech J Piotrowski (piotrow@toya.net.pl)
Zofia Kurmanowska (zofia_kurmanowska@tlen.pl)
Adam Antczak (adam.antczak1@wp.pl)
Jerzy Marczak (juraszka@wp.pl)
Pawel Górski (p_gorski@toya.net.pl)

Version: 3 Date: 3 April 2010

Author's response to reviews:

Dear Editor,

The following corrections have been made:

1. Author's contribution: the statement that all authors had read and approved the final version was added
2. Figure legend moved according to instructions
3. Tables moved according to instructions
4. Titles deleted
5. The title of a paper format was changed
6. Patients and methods changed to Methods
7. Abbreviations changed to a paragraph
8. Figures were rearranged

Best regards

W. Piotrowski